UNIVERSITY BIRMINGHAM University of Birmingham Research at Birmingham

Co-occurring psychotic and eating disorders in England

Rodgers, Ellen; Marwaha, S; Humpston, C

DOI: 10.1186/s40337-022-00664-0

License: Creative Commons: Attribution (CC BY)

Document Version Publisher's PDF, also known as Version of record

Citation for published version (Harvard):

Rodgers, E, Marwaha, S & Humpston, C 2022, 'Co-occurring psychotic and eating disorders in England: findings from the 2014 Adult Psychiatric Morbidity Survey.', *Journal of Eating Disorders*, vol. 10, no. 1, 150. https://doi.org/10.1186/s40337-022-00664-0

Link to publication on Research at Birmingham portal

General rights

Unless a licence is specified above, all rights (including copyright and moral rights) in this document are retained by the authors and/or the copyright holders. The express permission of the copyright holder must be obtained for any use of this material other than for purposes permitted by law.

•Users may freely distribute the URL that is used to identify this publication.

•Users may download and/or print one copy of the publication from the University of Birmingham research portal for the purpose of private study or non-commercial research.

•User may use extracts from the document in line with the concept of 'fair dealing' under the Copyright, Designs and Patents Act 1988 (?) •Users may not further distribute the material nor use it for the purposes of commercial gain.

Where a licence is displayed above, please note the terms and conditions of the licence govern your use of this document.

When citing, please reference the published version.

Take down policy

While the University of Birmingham exercises care and attention in making items available there are rare occasions when an item has been uploaded in error or has been deemed to be commercially or otherwise sensitive.

If you believe that this is the case for this document, please contact UBIRA@lists.bham.ac.uk providing details and we will remove access to the work immediately and investigate.

RESEARCH

Open Access

Journal of Eating Disorders

Co-occurring psychotic and eating disorders in England: findings from the 2014 Adult Psychiatric Morbidity Survey



Ellen Rodgers¹, Steven Marwaha^{1,2} and Clara Humpston^{1,3*}

Abstract

Background Psychotic disorders and eating disorders are complex mental illnesses associated with increased mortality and functional impairment. This study aimed to investigate the co-occurrence and relationships between eating disorders and psychotic disorders and assess the mediation effect of mood instability.

Methods This study used data from the Adult Psychiatric Morbidity Survey (APMS) 2014, a general populationbased survey in England. Participants (total N = 7546, female N = 4488, male N = 3058, mean age = 52.3 years) were categorised based on psychotic disorder status into the groups of probable psychosis, diagnosed psychosis, and healthy controls without psychosis. The dependent variable of this study was the presence or absence of an eating disorder, with mood instability as the mediator. Logistic regression and mediation analyses were conducted to assess the relationships between these variables.

Results Both probable and diagnosed psychoses were significantly related to the presence of an eating disorder, and mood instability was found to be a mediating variable with moderate effect.

Conclusion The present study demonstrates a significant relationship between eating disorders and psychotic disorders in the English general population, indicating higher levels of co-occurrence between these two groups of disorders than when compared with healthy controls. The findings also suggest the relationship between eating and psychotic disorders is mediated, to an extent, by the presence of mood instability traits. Future research could extend the present study's findings through assessing whether specific eating disorders are more significantly related to psychotic disorders than others.

Plain English summary

Eating disorders as well as psychotic disorders pose a significant risk to those diagnosed, with anorexia nervosa having the highest mortality rate among all mental disorders. The two groups of disorders are shown to be related as those with psychotic disorders are more likely to also be diagnosed with an eating disorder. Further, the relationship between eating and psychotic disorders could be due to trait mood instability. Mood instability is defined as the rapid switching between moods that those affected cannot control easily and has been found in individuals who are diagnosed with either an eating or a psychotic disorder. This leads to the question whether

*Correspondence: Clara Humpston clara.humpston@york.ac.uk

Full list of author information is available at the end of the article



[©] The Author(s) 2022. **Open Access** This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/licenses/by/4.0/. The Creative Commons Dedication waiver (http://creativecommons.org/publicdomain/zero/1.0/) applies to the data made available in this article, unless otherwise stated in a credit line to the data.

mood instability is involved in the relationship between psychotic and eating disorders. As individuals with both eating and psychotic disorders experience high levels of distress and treatment is challenging, it is important to understand the nature of the relationship between these two disorders.

This study used data collected from a population-based survey in England to examine whether psychotic disorders were related to eating disorders, and whether mood instability was implicated in the relationship. The study found that psychotic and eating disorders were significantly related, and mood instability had a moderate effect on the relationship.

Keywords Eating disorders, Psychotic disorders, Comorbidity, Mood instability, Epidemiology, Household survey

Background

Psychotic disorders (PD) and eating disorders (ED) are complex mental illnesses associated with increased mortality [1, 2] and poorer quality of life [3, 4]. EDs have become increasingly common over the past decade; in addition to the negative impact on the individual, anorexia nervosa holds the highest mortality rate across all psychiatric disorders [2]. Moreover, ED diagnosis is challenging due to the complexity of the disorders, which results in challenges with treatments too [2, 5].

The relationship between PD and ED remains unclear and not systematically well-studied [6, 7], although some evidence suggests the two groups of disorders are related to an extent [8, 9]. Much of the evidence supporting the link between ED and PD comes from case studies. Hugo and Lacey [8] report four cases of patients with co-occurring PD and anorexia nervosa (AN)/bulimia nervosa (BN). In two patients, psychotic symptoms worsened following improvement in ED symptoms, leading the authors to conclude that in these patients, disordered eating behaviours protected against psychotic symptoms [8]. Seeman [7] supports this, explaining that disordered eating offers psychological control. The control one may gain from restricting and changing eating behaviour is said to compensate for otherwise low self-efficacy in individuals at risk of psychosis, hence it is possible to extend Seeman's [7] hypothesis surrounding control and disordered eating to those with a PD. Researchers describe reduced feelings of agency in psychosis as 'passivity symptoms' [10] as those affected have issues with feelings and judgements of agency regarding their actions [11]. Hauser et al. [12] support this, stating that individuals who experience reduced cognitive and emotional control via psychosis use disordered eating to regain control. Consequently, PD and ED may be linked through the role of cognitive control.

In early accounts of EDs, the patients' disordered thinking styles were regarded as psychotic [13]. This idea is supported by recent empirical studies. In Miotto and colleagues' study, they assessed the level of psychotic symptoms in ED patients (with BN or AN) and compared them to a control group. ED patients reported more frequent experiences of psychotic, including paranoid, symptoms [9]. Similar experiences with psychotic symptoms have been reported in binge eating disorder (BED) patients. Aragona and colleagues found that increased psychotic experiences were linked with the severity of BED symptoms. Whilst findings from Aragona et al. [13] should be interpreted with caution due to the lack of a control group, other researchers support the existence of delusional thinking in EDs too. Behar, Arancibia, Gaete, Silva & Meza-Concha [14] found that individuals diagnosed with EDs showed delusional thinking patterns like those observed in PD. Delusions in EDs focus on body image and weight, suggesting that PD and ED may be linked through the shared symptom of delusional or delusionlike ideation [13, 14].

Another factor that may link the psychopathology of PD and ED is mood instability. Mood instability is defined as rapid oscillations in mood where individuals struggle to regulate their mood or the consequences associated with their behaviour [15]. Unstable mood is observed in many mental disorders, including PD [16] and ED [17]. As mood instability is associated with increased service use and suicidal ideation [18], it is important to consider the role unstable mood may have in co-occurring PD and ED.

At present, the evidence surrounding the role mood instability has in the relationship between PDs and EDs is sparse. A limited body of research suggests that those with AN or BN may be susceptible to psychotic symptoms and unstable mood [17]. As mood instability has been implicated in PD and ED individually [16, 19, 20], there is a possibility that the relationship between PD and ED may also be mediated by mood instability.

It is often thought that psychotic symptoms result from cognitive disturbances [21]. Cognitive distortions that lead to these symptoms are largely due to a breakdown in normal cognitive function, including the observation that individuals with PD often characterise self-generated cognitions as unfamiliar or coming from an external source [21]. From this, it is possible that the explanation for errors in source monitoring observed in PD are due to other, wider impairments in cognition. Fusar-Poli et al's [22] meta-analysis supports this, as individuals diagnosed with PD showed poorer executive function skills. This is extended by Aase et al. [23] who found that poor executive functioning resulted in increased positive symptoms of psychosis (such as hallucinations and delusions). Consequently, atypical cognitive function appears to have a role in PD [21–23]. As mentioned previously, low selfefficacy and cognitive control are observed in those with co-occurring PD and EDs [7]. Therefore, cognitive disturbances and poor cognitive control may also explain why PD and ED may be related.

Based on previous research highlighting the potentially significant relationship between the groups of PDs and EDs [8, 9, 13] and the possible role of mood instability in this relationship [17], the present study aimed to answer the following research questions:

- 1. Is the presence of psychotic symptoms, such as paranoid and hallucinatory experiences, associated with the presence of a diagnosed eating disorder?
- 2. Does the strength of association between psychotic symptoms and diagnosed eating disorder differ between groups of individuals who have diagnosed or probable psychosis, as compared with controls?
- 3. Does mood instability mediate the relationship between psychotic symptoms and diagnosed eating disorders?

Methods

Data collection and participants

The data used in this study were collected as part of the Adult Psychiatric Morbidity Survey (APMS) 2014. To select participants, the APMS 2014 employed a random, multi-stage stratified sampling method based on postcode sectors in England (of which there were 2550), ensuring an even spread across England. This was followed by another round of random sampling of privately owned household addresses within the sectors. From the selected households, one adult over the age of 16 was randomly selected for an interview resulting in 7546 participants (female N=4488, male N=3058, mean age=52.3 years) with responses to over 1000 variables in the final dataset. Participants were able to decline the invitation to participate as participation was voluntary.

Following participant consent, interviewers contacted participants for a semi-structured interview where multiple screens for psychiatric disorders were administered. The APMS 2014 used screening tools to assess common mental disorders, mental health treatment and service use, posttraumatic stress disorder, psychotic disorder, autism spectrum disorder, personality disorder, attention-deficit/hyperactivity disorder, bipolar disorder, alcohol dependence, suicidal thoughts, suicide attempts and self-harm. Depending on a participant's response determined whether they were contacted for an additional, second semi-structured interview. The APMS 2014 was designed to be representative of the entire population in England, hence the extensive sampling method and large sample size. A full, detailed breakdown of the APMS 2014 can be found in McManus, Bebbington, Jenkins & Brugha [24].

Psychotic symptom assessment

The Psychosis Screening Questionnaire (PSQ; [25]) was administered to participants during the first phase of APMS 2014 interviews. Participants were asked several questions used to indicate the presence of probable psychosis or a diagnosed PD. This also included asking participants whether they took antipsychotic medication, if they had been hospitalised for mental or emotional problems, whether they had a PD diagnosis, as well as a positive response to the PSQ items concerning experiences of auditory hallucinations and persecutory delusions.

The PSQ is comprised of five sections assessing different elements associated with psychotic experiences and PD: hypomanic mood, thought interference, paranoia, strange experiences, and auditory hallucinations. Participants gave yes/no/unsure answers to the questions, based on their experiences in the past year. The authors, Bebbington and Nayani, purposely made the first questions of each section broad so as not to overwhelm participants and to exclude sections if negative responses were given. The number of individuals who complete the PSQ and are identified as having potential PD is similar to that of those who would be clinically diagnosed using a standardised measure [26, 27], suggesting that the PSQ has good construct validity. From their original study, the PSQ also possesses very high levels of specificity (98 -100%) for use as a screening instrument.

Following the completion of the PSQ within the first phase, those who were identified as having probable psychosis or a PD were invited to participate in phase two, a second semi-structured interview. This equated to 6% of the original population. The Schedules for Clinical Assessment in Neuropsychiatry [28] was administered by a trained clinician. This process is reported in full by McManus et al. [24].

In the present study, individuals who answered yes to the PD diagnosis in the past 12 months by a medical professional from the variable (PDg12Doc_ReB) which in turn formed the diagnosed psychosis (clinical PD) group. This variable was chosen as it asks participants directly if they have been diagnosed with a PD or schizophrenia in the past 12 months by a medical professional. Those whoscreened positive in the probable PD (PsychDis-Prob) variable formed the probable PD group. Those who answered yes to this variable were identified as positive for psychotic symptoms following the SCAN but were not identified as having a formal diagnosis for PD or schizophrenia. All other individuals formed the control group.

ED assessment

The presence of an ED was indicated by responses to several questions on the APMS 2014. Participants were categorised as having an ED if they self-identified as having an ED when asked by the interviewer, had an ED diagnosis from a medical professional, or were receiving treatment for an ED at the time of the interview. The APMS 2014 also includes information on the age of onset of ED but does not include a specific questionnaire or screen to examine ED symptoms.

Both grouping variables for the clinical PD and probable PD groups were based on participant experiences in the past 12 months. Thus, for consistency, the present study uses the variable from the APMS 2014 dataset which asks participants whether they were diagnosed with an ED in the past 12 months by a medical professional (PDg9Ly_ReB). Although it is known that EDs include many varied disorders that are categorised by different symptoms, the present study is not able to differentiate between the different EDs and therefore cannot assess the extent to which the specific ED disorders are related to PD presence. This is due to the way the ED data was collected in the APMS. Thus, there will be just two levels to the binary ED variable: positive for diagnosed ED, or negative for diagnosed ED.

Mood instability

Mood instability was defined using the personality disorder scale, as administered during the APMS 2014. Participants who endorsed either of the PD79 or PD82 items were categorised as having unstable mood. The personality disorder scale is primarily designed to measure the presence of a personality disorder, yet the items selected for analysis in the present study are also associated with mood instability. PD79 asks if participants believe they are 'often doing things impulsively', whilst PD82 asks if participants feel they 'often have a lot of sudden mood changes'. These were both dichotomous variables with 'yes' or 'no' answers.

Data analysis

Data analysis was conducted using the programming language R in R studio and STATA SE/16. Analyses were weighted using the 'weight-core' variable in the APMS 2014 dataset. This accounts for non-response across the data to ensure results are representative of the general population. This also reduces the risk of over or underestimating the true prevalence of the disorders assessed in the APMS 2014. Any incomplete data due to withdrawal from the study or non-response was excluded from the final analysis.

Descriptive statistics were obtained based on sociodemographic features such as age, gender, ethnicity, socioeconomic status, and employment status for the clinical (diagnosed) PD, probable PD and control groups, with weighted Chi-squared tests carried out between diagnosed PD and probable PD groups accordingly.

We included three logistic regression models in the analyses: one assessing the relationship between probable PD and ED presence, one assessing the relationship between diagnosed PD and ED presence, and an alternative model including specific psychotic symptoms (delusions and hallucinations). Regression analyses were conducted in two stages, with sociodemographic variables that demonstrated significant differences in Chi-squared tests entered into stage 1 as explanatory variables, and various PD variables of interest were entered into stage 2 of the model as additional explanatory variables. ED presence was the dependent variable in all analyses. Weighted logistic regression analyses were performed to obtain odds ratios and 95% confidence intervals for ease of interpretation.

Lastly, mediation analyses assessed the effect the two mood instability variables (PD79 and PD82) had on the association between PD groups and ED presence. This measured the extent to which impulsive behaviour and unstable mood impacted the relationship between diagnosed PD, probable psychosis and ED. The KHB command in STATA was used to conduct this mediation analysis. This method of mediation analysis compares logistic regression models using the average partial effect [29].

Results

Descriptive statistics can be found in Table 1. The prevalence of diagnosed PD as diagnosed by a medical professional in the past year in the APMS 2014 survey was 0.6% (N=45), whilst the prevalence of probable PD was 1.2% (N=94).

In the diagnosed PD group, there was a higher proportion of males (60%), unemployed individuals (89%), people of white ethnicity (84%), and those living alone (78%). The greatest proportion of PD diagnoses was observed in the 35–44 age group (31%).

In the probable PD group, there were a higher proportion of females (59%), individuals in the 35–44 and 45–54 age groups (45%), those who were unemployed (89%), people of white ethnicity (89%), and those living alone (77%).

The first regression analysis revealed that ED presence in the past year was significantly associated with the co-occurrence of diagnosed PD (p<0.001, odds ratio (OR):12.92, 95% Confidence Interval (CI):3.36–49.70). In this model, age was found to be statistically significant (p=0.013, OR:0.84, CI:0.73–0.96), as was gender (p=0.023, OR:2.4, CI:1.14–5.08), and marital status (p<0.001, OR:0.25, CI:0.11–0.55).

Table 1 Descriptive statistics of the sample with results from weighted Chi-squared tests

	Diagnosed Psychosis (N=45)		Probable Psychosis (N = 94)			
	N (% of total)	Weighted χ^2	р	N (% of total)	Weighted χ^2	р
Gender		6.29	0.01		0.049	0.83
Male	27 (60)			39 (41.5)		
Female	18 (40)			55 (55.5)		
Employment		30.46	< 0.001		75.48	< 0.001
Employed	5 (11.1)			22 (10.6)		
Unemployed (including not economically active)	40 (88.9)			72 (89.4)		
Ethnicity		2.18	0.14		0.14	0.71
White	38 (84.4)			84 (89.4)		
Non-white	7 (15.6)			10 (10.6)		
Marital Status		11.88	< 0.001		27.05	< 0.001
With others	10 (22.2)			22 (10.6)		
Alone	35 (77.8)			72 (89.4)		
Age band		18.10	0.006		11.31	0.08
75+	2 (4.4)			3 (3.2)		
65–74	5 (11.1)			9 (5.7)		
55–64	9 (20.0)			17 (18.1)		
45–54	6 (13.3)			21 (22.3)		
35–44	14 (31.1)			21 (22.3)		
25–34	8 (17.8)			16 (17.0)		
16–24	1 (2.2)			7 (7.4)		

 Table 2
 Results from weighted logistic regression with eating disorder diagnosis as outcome (dependent) variable. OR, odds ratios; CI, confidence intervals

Measure	β (95% Cl)	Р	OR (95% CI)
Model 1			
Stage 1			
Age Gender Ethnicity Marital Status	-0.18 (-0.31-0.038) 0.88 (0.13-1.63) -0.66 (-1.41-0.084) -1.37 (-2.17 0.58)	0.012 0.022 0.081 < 0.001	0.83 (0.73–0.96) 2.40 (1.14–5.08) 0.51 (0.24–1.09) 0.25 (0.11–0.56)
Stage 2			
Diagnosed Psychosis	2.56 (1.21–3.91)	< 0.001	12.92 (3.36–49.70)
Model 2			
Stage 1			
Ethnicity Marital Status	-0.13 (-0.99–0.72) -1.42 (-2.27 0.57)	0.77 0.0011	0.88 (0.37–2.09) 0.24 (0.10–0.57)
Stage 2			
Probable Psychosis	3.35 (2.44–4.26)	< 0.001	28.42 (11.42–70.73)
Alt Model			
Stage 1			
Ethnicity Marital Status	-2.04 (-3.79–0.96) 1.30 (-3.10–0.49)	0.25 0.16	0.24 (0.023–2.61) 0.27 (0.045–1.64)
Stage 2			
Persecutory Delusions Auditory Hallucinations	-0.32 (-2.12–1.48) 0.20 (-1.82–2.21)	0.73 0.85	0.73 (0.12–4.39) 1.22 (0.16–9.12)

The second logistic regression revealed that ED presence in the past year was statistically significantly associated with probable PD (p<0.001, OR: 28.42, CI:11.42–70.73). In this model, only marital status was found to be significant (p=0.0011, OR:0.24, CI:0.10–0.57).

Due to the highly significant results from the two models, a third regression ('Alt model') was performed using the PSQ3a (persecutory delusions: felt that people were deliberately acting to harm you/your interests) and PSQ5a (auditory hallucinations: heard voices saying quite a few words or sentences) from the PSQ. These items aim to assess whether individuals have experienced clinically meaningful delusions or hallucinations typical of a PD, meaning that they would have also already endorsed PSQ3 and PSQ5 which ask about less specific psychoticlike experiences. However, the regression analyses performed for PSQ3a/PSQ5a and ED diagnosis did not produce statistically significant results (see Table 2).

Mediation analyses were conducted to assess the role mood instability had in the relationship between ED and diagnosed/probable PD. Both mood instability variables combined mediated the effect of diagnosed PD and ED by 18.31%. The variable 'often doing things impulsively' accounted for 8.22% of the effect diagnosed PD had on ED presence. Yet, the variable 'often have a lot of sudden mood changes' accounted for 15.36% of the effect diagnosed PD had on ED presence.

10.7% of the relationship between probable PD and ED was mediated by both mood instability variables. 5.69% of the effect probable PD had on ED presence was accounted for by the variable 'often doing things

 Table 3
 Results from mediation analyses with eating disorder diagnosis as dependent variable. OR, odds ratios; CI, confidence intervals. Asterisks indicate levels of statistical significance

	Probable Psychosis	Diagnosed Psychosis				
OR (95% CI) , <i>p</i> ^{***} (<0.001), <i>p</i> ^{**} (<0.01), <i>p</i> [*] (<0.05)						
Impulsivity + Mood Instability	10.70%	18.31%				
Total	12.86 (6.45–25.65) ***	6.23 (1.78–21.82) **				
Direct	9.79 (4.68–20.46) ***	4.46 (1.25–15.95) **				
Indirect	1.31 (1.04–1.66) **	1.40 (1.08–1.82) **				
Impulsivity	5.69%	8.22%				
Total	13.26 (6.75-26.06) ***	6.70 (1.94–23.15) **				
Direct	11.45 (5.80–22.58) ***	5.73 (1.66–19.84) **				
Indirect	1.16 (1.02–1.31) **	1.17 (1.00–1.37)				
Mood Instability	8.90%	15.36%				
Total	13.03 (6.47-26.2) ***	6.46 (1.89–22.08) **				
Direct	10.37 (4.94–21.78) ***	4.85 (1.39–16.87) **				
Indirect	1.26 (1.01-1.31) *	1.33 (1.07–1.66) **				

impulsively, whereas 8.90% of the effect probable PD had on ED presence was accounted for by the variable 'often have a lot of sudden mood changes.' For total, direct, and indirect effects of the mediator variables, see Table 3.

Discussion

From the above findings, we can tentatively suggest that there is a significant relationship between diagnosed PD and a diagnosis of an ED, and probable PD and a diagnosis of an ED. Such that, individuals with a diagnosed PD, or those in the probable PD group, are more likely to have also been diagnosed with an ED compared to controls. Mood instability was found to account for a small to moderate amount of the association between diagnosed PD and ED, with emotional instability as opposed to behavioural impulsivity being the strongest mediator variable. Mood instability was found to have a smaller mediation effect in the relationship between probable PD and ED, but emotional instability was the strongest mediator in this model too. Sociodemographic information, including age, gender, employment status and marital status, also demonstrated statistically significant differences between the PD groups; however, even after accounting for these sociodemographic variables in the regression models, the relationship between the PD groups and an ED diagnosis persisted.

When discussing the symptoms and experiences of individuals diagnosed with ED, researchers highlight similarities with PD. Those with AN are described as having delusional thinking patterns surrounding their body image and weight [14, 30]. Behar et al., [14] state that delusional beliefs in EDs exist on a spectrum from over-valuing body image to overt delusion surrounding one's body and size [30]. These delusions in EDs, it is said, encompass the denial of one's own diagnosis [14]. Denial or lack of insight, along with the finding that weight loss can exacerbate psychotic symptoms, such as hallucinations and delusions, [8, 14] suggest that there is potentially a psychotic element to EDs, supporting the idea that PD and ED are related.

Mood instability has been found to impact ED development and symptoms as well. Brown, Hochman and Micali [20] investigated the role mood instability during childhood played in ED development. Emotional instability and depressed mood were both found to be significant predictors of disordered eating behaviour at age 14 and 16. This is a particularly important finding surrounding the influence mood instability has on ED development. Low mood and depressive symptoms have been linked to the occurrence of binge/purge behaviour in BN patients [31] as well as heightened levels of impulsive behaviour [32]. Thus, it could be argued that unstable mood, particularly unstable negative moods, are significantly associated with EDs [20, 31, 32].

Further research from Miniati et al., [17] suggests that mood instability is found in patients with AN or BN. Additionally, BN patients were more prone to psychotic symptoms and depressive moods compared to AN patients. Therefore, these findings could be interpreted as linking psychosis, ED, and mood instability. This, along with the fact that mood instability increases the risk for suicide in both PD and ED patients [17, 19] mean that it is important to further investigate the role mood instability plays in the relationship between ED and PD.

A possible explanation for the findings in the present study may be genetics. It has been shown that respectively, EDs and PDs are moderately heritable disorders [33, 34], that may be related at a genetic level [35]. Findings from Zhang et al. [35] suggest that in families where an individual has a diagnosed ED, their relatives have a greater risk of being diagnosed with schizophrenia than control families. It has also been found that schizophrenia and mood instability have a shared genetic basis, which may help to explain the role mood instability had as a mediator between ED and PD in the present study [36]. Hindley et al. [36] emphasise the findings from their study should be interpreted with caution as the correlation between gene loci in mood instability and schizophrenia is weak to moderate. However, these findings could be seen to support the results observed in the present study when used alongside those by Zhang et al.: that EDs and PDs are significantly related and mood instability acts as a mediator, due to shared genes between the disorders which may go beyond EDs and PDs and extend into other mental health disorders and even physical health conditions. The co-occurrence of EDs and PDs could indeed be viewed through the lens of comorbid disorders beyond these categories, for example, the mediating role of mood instability may also at least partly explain the high rates of depression and anxiety

associated with ED diagnoses. However, good quality evidence on comorbid PD and ED is scarce compared to data assessing mood disorders alongside EDs.

The finding that PD and ED are associated with one another is in line with previous research [8, 9, 13] and is valuable as the body of evidence is relatively small [6]. The APMS 2014 dataset used in this secondary analysis was large in comparison to sample sizes of other studies. This study involved over 7000 participants, therefore the findings from this study could be widely generalisable.

The present study involved the use of a large dataset (N=7546), which may be one of the largest cohorts used to assess the relationship between PD and ED. This is a strength as one of the largest studies prior to the present study included 112 participants [9], and other studies are small case studies. Additionally, the method in which the APMS 2014 data was collected means the prevalence of co-occurring PD and ED is likely to be representative of the general population of England.

This study, to our knowledge, is one of the first to assess the role of mood instability in relation to co-occurring PD and ED. Previous research from Miniati et al., [17] seemed to suggest that individuals with ED were more susceptible to mood instability and psychotic symptoms. However, Miniati and colleagues' participants were not formally diagnosed with a PD. Consequently, the present study extends Miniati et al.'s [17] findings to those with formal diagnoses of both ED and PD, as well as those with probable PD. These findings can also be seen to support research that highlights the significant role of mood instability in ED [20, 32] and PD [16].

Moreover, the items used in the present study to establish mood instability can be found in a scale used by the APMS 2014 to measure personality disorder. Whilst most of the personality disorder scale looks at state-level mood disruption, the mediator variables used in the present study focus on trait mood instability. A previous longitudinal study from Brown, Hochman & Micali [20] emphasises that trait emotional instability is associated with the development of ED. This too was found in Ghaderi & Scott [37] as individuals with ED scored lower on trait emotional stability variables have been shown to be most applicable to ED, this can be seen as a strength of the present study.

This study, however, has several limitations. As the APMS 2014 collected data from individuals living in private households in England, this means that individuals living in sheltered accommodation, individuals hospitalised at the time of data collection or homeless populations would not have been included in this study. A meta-analysis conducted by Ayano, Tesfaw and Shumet [38] found that the prevalence of PDs in homeless people (21.21%) was much greater, compared to the general

population (12.7%). Similar results were found when comparing the prevalence of ED diagnosis in populations of homeless versus housed US Veterans. It has been demonstrated that homelessness increases the probability of developing an ED by 59% [39]. This suggests that the present study may under-estimate the true prevalence of both PDs and EDs as the study's population does not include populations where the disorders are more prevalent [38, 39].

Another limitation is that the number of individuals in each of the PD groups was relatively small, the association between diagnosed/probable PD and ED therefore may be inflated, meaning findings should be interpreted with caution. In the APMS 2014, the data containing information on ED diagnoses is based on answers given to yes or no questions, meaning the variable is binary. It is argued that ED should be considered as dimensional [40, 41], as other authors note that disordered eating behaviour ranges from occasional restrictive diets to fullblown EDs [40]. Additionally, compared to a categorical approach of ED as well PD, dimensional approaches enhance classification and are more sensitive to comorbid symptoms such as impulsivity [41]. Therefore, the categorical approach adopted by the APMS 2014 can be criticised for not being sensitive enough to the differing levels of ED severity [40, 41].

An issue with the present study is that all ED are grouped into one variable. As noted by the DSM-5, there are four main ED categories: AN, BN, BED, and other specified feeding and eating disorder [42]. The main categories of ED diagnoses can be split further into more specific diagnoses which all differ. For example, individuals with purgative BN display a more negative emotional profile compared to non-purgative BN and non-specific BN [43]. This is supported by Anderson et al., [44] who found that AN-binge-purge patients and BN patients reported more difficulties with emotional regulation. Due to the differences observed across the ED diagnoses, it is possible to argue the grouping of ED patients into one group is not entirely appropriate. Consequently, the APMS 2014 is not able to distinguish what specific ED are more significantly associated with which PDs.

The same can be said for the PD variable used in the present study as there is no way to distinguish the different diagnoses from one another. The PDs, including schizophrenia, psychotic depression, and bipolar disorder, all have interrelated yet different cognitive profiles. Research suggests that the degree of cognitive impairment is worst in schizophrenia [45, 46]. Differences between PDs are also observed at a neurobiological level, as reported by Culbreth, Foti, Barch, Hajcak & Kotiv [47]. It was found that those with an affective psychosis showed the greatest brain activity in response to emotional stimuli. From this, it can be argued there

are distinct differences between the PDs [45, 47] which are not detected in the present study. Therefore, it cannot be said whether a particular PD is more frequently observed in those with ED, due to the way participants were categorised.

Lastly, and perhaps the most important limitation of the present study, is the fact that the categorisation of the participants into groups (positive for ED, negative for ED, clinical PD, probable PD and controls) was reliant on self-report. Whilst the study makes use of the variables obtained by the APMS where participants were asked to declare if they were diagnosed with an ED or PD in the past 12 months by a medical professional, it is possible some responses may have been untrue. Particularly in ED it has been shown that patients often lie about the severity of their disordered eating, preferring to minimise the severity of their symptoms [48]. Therefore, the reliance on self-reported diagnoses in the present study may have resulted in potentially inaccurate estimates of the prevalence of PD, ED and ultimately the extent to which they are related. Hence the findings from the present study should be interpreted cautiously due to this limitation.

To address the limitations of the present study, future research could administer an ED questionnaire to specify which EDs are the most significantly associated with psychotic symptoms. Possible questionnaires are the Eating Disorder Inventory (EDI; [49]), the Eating Disorder Examination Questionnaire (EDE-Q; [50]) and the Eating Attitudes Test (EAT; [51]). The EAT is perhaps one of the most widely used psychometric tools for assessing EDs and has even been shortened to the 16-item version [51]. Despite its widespread use, the EAT is thought of as AN-specific [52], meaning the EDI may be best suited to comparing a range of EDs of differing severities. The EDI has been identified as having good psychometric properties, including good internal consistency [53, 54]. The EDE-Q has been shown to produce similar findings to that of the Eating Disorder Examination (EDE; [50]), the investigator-based interview the EDE-Q is based on. The EDE-Q has been shown to have good internal consistency and other psychometric properties [55]. Yet, a short version known as the EDE-Q7 [56, 57] is described as offering more robust psychometric properties than the original version [58].

Additionally, future research could aim to assess what specific PD are associated with ED, whilst also accounting for PD severity as seen in the present study. The SCAN [28] is a widely used method for assessing PD and can be used to specify which PD an individual has [59]. Although the SCAN was used in the present study, the APMS 2014 did not specify the prevalence of specific PD. Future research could also use the SCAN, but group individuals based on their specific PD and severity of their symptoms, thus allowing for comparisons to be made between different PD regarding the strength of their relationships with ED. This would develop a better understanding into the potentially psychotic elements in at least some ED [13, 14] which would in turn, improve treatment options for those individuals too.

Lastly, mood instability and its role in the relationship between PD and ED should be investigated further as this mediator variable is not well studied in this area of research. Although findings from Miniati et al., [17] and the present study suggest mood instability does mediate the relationship between PD and ED at least to a certain degree, further studies are clearly warranted including those adopting a longitudinal design.

Conclusion

Overall, the present study supports previous research that PD and ED do co-occur in the general population. Despite the present study's limitations, its findings contribute additionally to the relatively small area of literature. Our findings also suggest that mood instability has a moderately mediating role in the relationships between the two disorders.

Abbreviations

APMS Adult Psychiatric Morbidity Survey EAT Eating Attitudes Test FD Eating Disorder FDI Eating Disorder Inventory FDF-O Eating Disorder Examination Questionnaire PD Psychotic Disorder SCAN Schedules for Clinical Assessment in Neuropsychiatry

Acknowledgements

We would like to thank the participants who took part in and gave their valuable time in the Adult Psychiatric Morbidity Survey 2014.

Authors' contributions

E.R. wrote the main manuscript text. E.R. and C.H carried out the analyses and prepared the figures. All authors reviewed the manuscript.

Funding

No external funding was received for this work.

Data availability

Data from the APMS can be officially obtained from the UK's NHS Digital upon reasonable request.

Declarations

Ethics approval and consent to participate

Not applicable - secondary data analysis only.

Consent for publication We consent for publication.

Competing interests

SM has received funding from Lundbeck, Sunovion and Janssen to attend educational events. Other authors declare no conflicts.

Author details

¹Institute for Mental Health, School of Psychology, University of Birmingham, B15 2TT Birmingham, UK ²Birmingham and Solihull Mental Health Foundation Trust, B1 3RB Birmingham, UK

³Department of Psychology, University of York, YO10 5DD York, UK

Received: 13 June 2022 / Accepted: 18 September 2022 Published online: 18 October 2022

References

- 1. Dutta R, Murray RM, Allardyce J, Jones PB, Boydell JE. Mortality in first-contact psychosis patients in the UK: a cohort study. Psychol Med. 2012;42(8):1649–61. https://doi.org/10.1017/S0033291711002807.
- Micali N, Hagberg KW, Petersen I, Treasure JL. (2013). The incidence of eating disorders in the UK in 2000–2009: findings from the General Practice Research Database. BMJ open, 3(5). https://doi.org/10.1136/ bmjopen-2013-002646.
- Arciniegas DB. (2015). Psychosis. Continuum: Lifelong Learning in Neurology, 21(3) Behavioral Neurology and Neuropsychiatry, 715. https://doi. org/10.1212/01.CON.0000466662.89908.e7.
- Rutter-Eley EL, James MK, Jenkins PE. Eating Disorders, Perfectionism, and Quality of Life: Maladaptive Perfectionism as a Mediator Between Symptoms of Disordered Eating and Quality of Life. J Nerv Ment Dis. 2020;208(10):771–6. https://doi.org/10.1097/NMD.00000000001241.
- Ciszewski S, Flood KE, Proctor CJ, Best LA. Exploring the Relationship Between Disordered Eating and Executive Function in a Non-Clinical Sample. Percept Mot Skills. 2020;127(6):1033–50. https://doi.org/10.1177/0031512520937569.
- Pereira CA, Silva JR, Cajão R, Lourenço JP, Casanova T. The co-occurrence of eating disorders and psychosis. Eur Psychiatry. 2016;33(S1):425–5. https://doi. org/10.1016/j.eurpsy.2016.01.1540.
- Seeman MV. Eating disorders and psychosis: seven hypotheses. World J psychiatry. 2014;4(4):112. https://doi.org/10.5498/wjp.v4.i4.112.
- Hugo PJ, Lacey JH. Disordered eating: a defense against psychosis? Int J Eat Disord. 1998;24(3):329–33. https://doi.org/10.1002/ (SICI)1098-108X(199811)24:3%3C329::AID-EAT12%3E3.0.CO;2-R.
- Miotto P, Pollini B, Restaneo A, Favaretto G, Sisti D, Rocchi MB, Preti A. Symptoms of psychosis in anorexia and bulimia nervosa. Psychiatry Res. 2010;175(3):237–43. https://doi.org/10.1016/j.psychres.2009.03.011.
- Waters FA, Badcock JC. First-rank symptoms in schizophrenia: reexamining mechanisms of self-recognition. Schizophr Bull. 2010;36(3):510–7. https://doi. org/10.1093/schbul/sbn112.
- 11. Leptourgos P, Corlett PR. Embodied predictions, agency, and psychosis. Front big Data. 2020;3:27. https://doi.org/10.3389/fdata.2020.00027.
- Hauser M, Knoblich G, Repp BH, Lautenschlager M, Gallinat J, Heinz A, Voss M. Altered sense of agency in schizophrenia and the putative psychotic prodrome. Psychiatry Res. 2011;186(2–3):170–6. https://doi.org/10.1016/j. psychres.2010.08.003.
- Aragona M, Petta AM, Balbi A. Psychotic phenomena in Binge Eating Disorder: an exploratory MMPI-2 study. Arch Psychiatry Psychother. 2015;17(2):13– 20. https://doi.org/10.12740/APP/43321.
- Behar R, Arancibia M, Gaete MI, Silva H, Meza-Concha N. The delusional dimension of anorexia nervosa: phenomenological, neurobiological and clinical perspectives. Archives of Clinical Psychiatry (São Paulo). 2018;45:15– 21. https://doi.org/10.1590/0101-60830000000148.
- Marwaha S, Broome MR, Bebbington PE, Kuipers E, Freeman D. Mood instability and psychosis: analyses of British national survey data. Schizophr Bull. 2014;40(2):269–77. https://doi.org/10.1093/schbul/sbt149.
- Marwaha S, He Z, Broome M, Singh SP, Scott J, Eyden J, Wolke D. How is affective instability defined and measured? A systematic review. Psychol Med. 2014;44(9):1793. https://doi.org/10.1017/S0033291713002407.
- 17. Eating and Weight Disorders-Studies on Anorexia, Bulimia and Obesity, 23(3), 305–311. https://dx.doi.org/10.1007/s40519-016-0333-1.
- Broome MR, Saunders KEA, Harrison PJ, Marwaha S. Mood instability: significance, definition and measurement. Br J Psychiatry. 2015;207(4):283–5. https://doi.org/10.1192/bjp.bp.114.158543.
- McGinty J, Haque MS, Upthegrove R. Depression during first episode psychosis and subsequent suicide risk: a systematic review and meta-analysis of longitudinal studies. Schizophr Res. 2018;195:58–66. https://doi.org/10.1016/j. schres.2017.09.040.
- Brown M, Hochman A, Micali N. Emotional instability as a trait risk factor for eating disorder behaviors in adolescents: Sex differences in a large-scale prospective study. Psychol Med. 2020;50(11):1783–94. https://doi.org/10.1017/ S0033291719001818.

- 21. Garety PA, Kuipers E, Fowler D, Freeman D, Bebbington PE. A cognitive model of the positive symptoms of psychosis. Psychol Med. 2001;31(2):189–95. https://doi.org/10.1017/S0033291701003312.
- 22. Archives of general psychiatry, 69(6), 562–571. https://dx.doi.org/10.1001/ archgenpsychiatry.2011.1592.
- 23. Aase I, Langeveld JH, Johannessen JO, Joa I, Dalen I, ten Velden Hegelstad W. Cognitive predictors of longitudinal positive symptom course in clinical high risk for psychosis. Schizophrenia Research: Cognition. 2021;26:100210. https://doi.org/10.1016/j.scog.2021.100210.
- 24. McManus S, Bebbington PE, Jenkins R, Brugha T. (2016). *Mental Health and Wellbeing in England: The Adult Psychiatric Morbidity Survey 2014*. NHS digital.
- Bebbington P, Nayani T. The Psychosis Screening Questionnaire. Int J Methods Psychiatr Res. 1995;5(1):11–9.
- 26. The British Journal of Psychiatry, 185(4), 298–305. https://dx.doi.org/10.1192/ bjp.185.4.298.
- Wiles NJ, Zammit S, Bebbington P, Singleton N, Meltzer H, Lewis G. Selfreported psychotic symptoms in the general population: results from the longitudinal study of the British National Psychiatric Morbidity Survey. Br J Psychiatry. 2006;188(6):519–26. https://doi.org/10.1192/bjp.bp.105.012179.
- Archives of general psychiatry, 47(6), 589–593. https://dx.doi.org/10.1001/ archpsyc.1990.01810180089012.
- Karlson KB, Holm A, Breen R. Comparing Regression Coefficients Between Same-sample Nested Models Using Logit and Probit: A New Method. Sociol Methodol. 2012;42(1):286–313. https://doi.org/10.1177/0081175012444861.
- Gadsby S. Self-deception and the second factor: How desire causes delusion in anorexia nervosa. Erkenntnis. 2020;85(3):609–26. https://doi.org/10.1007/ s10670-018-0039-z.
- Berner LA, Crosby RD, Cao L, Engel SG, Lavender JM, Mitchell JE, Wonderlich SA. Temporal associations between affective instability and dysregulated eating behavior in bulimia nervosa. J Psychiatr Res. 2017;92:183–90. https:// doi.org/10.1016/j.jpsychires.2017.04.009.
- Cooper M, Todd G. Selective processing of three types of stimuli in eating disorders. Br J Clin Psychol. 1997;36(2):279–81. https://doi. org/10.1111/j.2044-8260.1997.tb01413.x.
- Biological psychiatry, 83(6), 492–498. https://dx.doi.org/10.1016/j. biopsych.2017.08.017.
- Klump KL, Suisman JL, Burt SA, McGue M, Iacono WG. Genetic and environmental influences on disordered eating: An adoption study. J Abnorm Psychol. 2009;118(4):797. https://doi.org/10.1037/a0017204.
- Molecular Psychiatry, 26(9), 5389–5397. https://dx.doi.org/10.1038/ s41380-020-0749-x.
- American Journal of Medical Genetics Part B: Neuropsychiatric Genetics. https:// dx.doi.org/10.1002/ajmg.b.32907.
- Ghaderi A, Scott B. (2000). The big five and eating disorders: A prospective study in the general population. Eur J Pers, 14(4), 311–23. https://doi. org/10.1002/1099-0984(200007/08)14:4%3C311::AID-PER378%3E3.0.CO;2-8.
- Ayano G, Tesfaw G, Shumet S. The prevalence of schizophrenia and other psychotic disorders among homeless people: a systematic review and meta-analysis. BMC Psychiatry. 2019;19(1):1–14. https://doi.org/10.1186/ s12888-019-2361-7.
- Livingston WS, Brignone E, Fargo JD, Gundlapalli AV, Maguen S, Blais RK. VHA-enrolled homeless veterans are at increased risk for eating disorder diagnoses. Mil Med. 2019;184(1–2):e71–5. https://doi.org/10.1093/milmed/ usy175.
- 40. Patton GC. The spectrum of eating disorder in adolescence. J Psychosom Res. 1988;32(6):579–84. https://doi.org/10.1016/0022-3999(88)90006-2.
- Wildes JE, Marcus MD. Incorporating dimensions into the classification of eating disorders: three models and their implications for research and clinical practice. Int J Eat Disord. 2013;46(5):396–403. https://doi.org/10.1002/ eat.22091.
- 42. American Psychiatric Association. (2013). Eating Disorders. In *Diagnostic and statistical manual of mental disorders* (5th ed.).
- Pascual A, Etxebarria I, Cruz MS. Emotional differences between women with different types of eating disorders. Int J Psychol Psychol Therapy. 2011;11(1):1–11.
- 44. Journal of clinical psychology, 74(10), 1867–1873. https://dx.doi.org/10.1002/ jclp.22638.
- Sheffield JM, Karcher NR, Barch DM. Cognitive deficits in psychotic disorders: a lifespan perspective. Neuropsychol Rev. 2018;28(4):509–33. https://doi. org/10.1007/s11065-018-9388-2.

- Culbreth AJ, Foti D, Barch DM, Hajcak G, Kotov R. (2018). Electrocortical responses to emotional stimuli in psychotic disorders: Comparing schizophrenia spectrum disorders and affective psychosis. Front Psychiatry, 586. https://doi.org/10.3389/fpsyt.2018.00586.
- Ali, K., Fassnacht, D. B., Farrer, L., Rieger, E., Feldhege, J., Moessner, M., ... Bauer,S. (2020). What prevents young adults from seeking help? Barriers toward help-seeking for eating disorder symptomatology. *International Journal of Eating Disorders*, 53(6), 894–906. https://dx.doi.org/10.1002/eat.23266.
- Garner DM, Olmstead MP, Polivy J. Development and validation of a multidimensional eating disorder inventory for anorexia nervosa and bulimia. Int J Eat Disord. 1983;2(2):15–34. https://doi.org/10.1002/1098-108X(198321)2:2%3C15::AID-EAT2260020203%3E3.0.CO;2-6.
- Fairburn CG, Beglin SJ. Assessment of eating disorders: Interview or self-report questionnaire? Int J Eat Disord. 1994;16(4):363–70. https://doi. org/10.1002/1098-108X(199412)16:4%3C363::AID-EAT2260160405%3E3.0 .CO;2-%23.
- Garner DM, Garfinkel PE. The Eating Attitudes Test: An index of the symptoms of anorexia nervosa. Psychol Med. 1979;9(2):273–9. https://doi.org/10.1017/ S0033291700030762.
- Ocker LB, Lam ET, Jensen BE, Zhang JJ. Psychometric properties of the eating attitudes test. Meas Phys Educ Exerc Sci. 2007;11(1):25–48. https://doi. org/10.1080/10913670709337010.
- Raciti MC, Norcross JC. The EAT and EDI: Screening, interrelationships, and psychometrics. Int J Eat Disord. 1987;6(4):579–86. https://doi.org/10.1002/1098-108X(198707)6:4%3C579::AID-EAT2260060418%3E3.0.CO;2-C.

- Schaefer WK, Maclennan RN, Yaholnitsky-Smith SA, Stover ED. Psychometric evaluation of the Eating Disorder Inventory (EDI) in a clinical group. Psychol Health. 1998;13(5):873–81. https://doi.org/10.1080/08870449808407437.
- International Journal of Eating Disorders, 40(4), 386–389. https://dx.doi. org/10.1002/eat.20373.
- Grilo CM, Henderson KE, Bell RL, Crosby RD. Eating disorder examinationquestionnaire factor structure and construct validity in bariatric surgery candidates. Obes Surg. 2013;23(5):657–62. https://doi.org/10.1007/ s11695-012-0840-8.
- Grilo CM, Reas DL, Hopwood CJ, Crosby RD. Factor structure and construct validity of the eating disorder examination-questionnaire in college students: Further support for a modified brief version. Int J Eat Disord. 2015;48(3):284– 9. https://doi.org/10.1002/eat.22358.
- Jenkins PE, Rienecke RD. Structural validity of the eating disorder examination—Questionnaire: A systematic review. Int J Eat Disord. 2022. https://doi. org/10.1002/eat.23721.
- Seiler N, Nguyen T, Yung A, O'Donoghue B. Terminology and assessment tools of psychosis: A systematic narrative review. J Neuropsychiatry Clin Neurosci. 2020;74(4):226–46. https://doi.org/10.1111/pcn.12966.

Publisher's Note

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.